Author's response to reviews

Title: Correlation between Progetto Cuore Risk Score and Early Cardiovascular Damage in Never Treated Subjects

Authors:

Aldo Pende (apende@unige.it)
Caterina Grondona (categrondona@virgilio.it)
Stefano Bertolini (stefbert@unige.it)

Version: 2 Date: 10 September 2008

Author's response to reviews: see over
To Eugenio Picano  
Editor-in-Chief  
Cardiovascular Ultrasound  

Please find herewith enclosed a revised version of our manuscript entitled “Correlation between Progetto Cuore Risk Score and Early Cardiovascular Damage in Never Treated Subjects”.  

We changed the text in agreement with the comments of the referees. In particular:  

Referee 1:  
In the Discussion Section (penultimate paragraph) we considered the issue raised by the referee about the results of our correlations and the prognostic performances shown by intima-media thickness and left ventricular mass. We think that the characteristics of our subjects explain the apparent contradiction of our results.  

We also discussed the reasons of some poor correlations (last paragraph).  

We prepared a figure with the graphs of the most significant correlations.  

Referee 2:  
No specific comments.  

Best regards, yours sincerely  

Aldo Pende, M.D.  
Department of Internal Medicine  
University of Genoa School of Medicine  
Viale Benedetto XV, 6  
16132 Genoa, Italy  
Tel. & Fax 00390103537524  
e-mail: apende@unige.it